Author’s response to reviews

Title: A "Grantathon" Model to Mentor New Investigators in Mental Health Research

Authors:

Mary Hawk (meh96@pitt.edu)
Vishwajit Nimgaonkar (vishwajitnl@upmc.edu)
Triptish Bhatia (bhatiatriptish@yahoo.co.in)
Jaspreet Brar (brarjs@upmc.edu)
Wafaa Abdelhakim Elbahaey (wafaaelbahaey@mans.edu.eg)
James Egan (jee48@pitt.edu)
Prasad Konsale (prasadkm@upmc.edu)
Supriya Kumar (supriya@pitt.edu)
Margaret McDonald (mmaggie@pitt.edu)
Ravinder Singh (aaressjay@gmail.com)
Soumya Swaminathan (soumya.s@nic.in)
Joel Wood (woodja@upmc.edu)
Smita Deshpande (smitadeshp@gmail.com)

Version: 1 Date: 02 Aug 2017

Author’s response to reviews:

Dr. Gonzalez-Mcquire,

Thank you for your interest in our paper as a commentary rather than a research article. We have made the necessary revisions for consideration as an article of this type and are happy to resubmit it at this time. We look forward to responses from peer review.
Best,

Mary Hawk